Author's response to reviews

Title: A Case of Focal Segmental Glomerulosclerosis in an Adult Patient with Hypogammaglobulinemia Superimposed on Membranoproliferative Glomerulonephritis in Childhood

Authors:

Kenji Tsuji (gmd422036@s.okayama-u.ac.jp)
Haruhito A Uchida (hauchida@md.okayama-u.ac.jp)
Tetsuichiro Ono (tym_ono_424@train.ocn.ne.jp)
Tasuyuki Inoue (ta_nyan2000@yahoo.co.jp)
Katsuji Shinagawa (sinagawa@md.okayama-u.ac.jp)
Shinji Kitamura (s-kita@md.okayama-u.ac.jp)
Yohei Maeshima (ymaeshim@md.okayama-u.ac.jp)
Hitoshi Sugiyama (hitoshis@md.okayama-u.ac.jp)
Hirofumi Makino (makino@md.okayama-u.ac.jp)

Version: 2 Date: 26 May 2012

Author's response to reviews: see over
May 26, 2012

Editor-in-Chief of BMC Nephrology
Prof. Hayley Henderson,

Please find attached our revised, original manuscript entitled “A Case of Focal Segmental Glomerulosclerosis in an Adult Patient with Hypogammaglobulinemia Superimposed on Membranoproliferative Glomerulonephritis in Childhood”, by Tsuji K et al., for publication in BMC Nephrology. We made answers to the Reviewers’ comment point by point, then modified our manuscript.

Patients with hypogammaglobulinemia seem to be less susceptible to immune complex and complement related nephritis. There are only 3 young cases in Pubmed as far as we know. Our adult case in this manuscript is the first case in this sense. Thus we would consider that our manuscript is worth publishing in your journal.

By submitting the manuscript to BMC Nephrology, the authors understand that the material presented in this paper has not been published before nor has it been submitted for publication to another scientific journal or being considered for publication elsewhere. I attest that this work has been approved by all co-authors. The authors also understand that should the submitted material be accepted for publication in the journal, they will automatically transfer the copyright to the publisher. No potential conflict of interest was disclosed.

I hope that our responses and changes to the manuscript have adequately addressed the reviewers’ concerns and the reviewing process finds this revised manuscript acceptable for publication in BMC Nephrology.

Sincerely Yours,

Haruhito Adam Uchida, M.D., PhD.
Department of Medicine and Clinical Science,
Okayama University Graduate School of Medicine, Dentistry and Pharmaceutical Sciences
2-5-1 Shikata-cho, Kita-ku, Okayama 700-8558, Japan
Phone: (+81) 86-235-7235
Fax: (+81) 86-222-5214
e-mail: hauchida@md.okayama-u.ac.jp